Author’s response to reviews

Title: Teriparatide use during an economic crisis: baseline data from the Greek cohort of the Extended Forsteo Observational Study (ExFOS)

Authors:

Kyriakos Aloumanis (Aloumanis_kyriakos@lilly.com)
George Kapetanos (geo.kapetanos@gmail.com)
Nikolaos Bartzis (bartzis_nikolaos@lilly.com)
Vangelis Drossinos (drossinos_vangelis@lilly.com)

Version: 2 Date: 30 June 2014

Author’s response to reviews:

An additional paragraph listing the Hellenic ExFOS Study Group has been added to the acknowledgements section.